Author's response to reviews

Title: Teaching evidence based medicine literature searching skills to medical students during the clinical years - a protocol for a randomised controlled trial

Authors:

    Dragan Ilic (dragan.ilic@monash.edu)
    Katrina Tepper (katrina.tepper@monash.edu)
    Marie Misso (marie.misso@monash.edu)

Version: 5 Date: 27 July 2011

Author's response to reviews:

Dear Editors,

Many thanks for accepting our protocol for publication. We have uploaded our manuscript according to the formatting changes requested;

- Abstract: Please change methods to methods/design on submission system. This has been amended as requested.
- Highlighting/tracking: Please remove from the manuscript. The highlighted text has been removed.

Best wishes,

Dragan Ilic